Author's response to reviews

Title: The quantity, nature and quality of Aboriginal and Torres Strait Islander Australian mentoring literature: A systematic review

Authors:

Roxanne Bainbridge (roxanne.bainbridge@jcu.edu.au)
Komla Tsey (komla.tsey@jcu.edu.au)
Janya McCalman (janya.mccalman@jcu.edu.au)
Simon Towle (Simon.Towle@datima.qld.gov.au)

Version: 2 Date: 16 September 2013

Author's response to reviews:

16 September 2013

Dear Proel,

Profuse apologies for lack of formatting in submitting the previous copy of the paper; I inadvertently attached the pre-formatted version. However, I attach this copy and have double-checked to ensure, to the best of my ability, that:

# There is double line spacing;
# The text is unjustified;
# There are no hyphenated words at line breaks;
# Hard returns have only been used to end headings and paragraphs;
# Only first words and proper nouns are capitalised in titles.
# All pages should be numbered;
# The BMC Public Health reference format has been used;
# There are no footnotes;
# The text is not formatted in multiple columns; and
# Tables and figures are submitted as attachments and not the main text.

Kind regards,

Roxanne Bainbridge